
SOHO Daily Meeting Minutes for Tuesday, 16 October 2001

ANNOUNCEMENTS
-------------
* There will be no EOF planning meetings until Friday.

FOT REPORT
----------
Spacecraft Status:       Nominal

Spacecraft Anomalies:    None

Accomplished Activities: VIRGO, SWAN, New Tracking Star

Planned Activities:      RSL, VIRGO, Extra VC2

Upcoming Operations:     None

Ground Anomalies:        288/1355  Wayward command echo block. D66.

INSTRUMENT STATUS
-----------------
CDS:    Nominal.  See MEDOC report.

SUMER:  Nominal.  See MEDOC report.

UVCS:   Nominal.  See MEDOC report.

LASCO:  Nominal.  C2 and C3 synoptics.

EIT:    Nominal.  Half resolution 195 Angstrom CME watch and synoptics. 
        EIT will do 45 minutes of high cadence observations today and
        tomorrow in support of JOP137.  Also supporting JOP104.

MDI:    Nominal.  High resolution observations.

TRACE:  Nominal.  Observing AR 9661 and supporting JOP104 and JOP137.

MEDOC Daily planning for Wednesday 17 October 2001:
---------------------------------------------------
Data shown from 15th for JOP 137 with good loops

Data received from JOP 104. No particular activity observed.

SUMER
slit set at x=990
0200-1800 JOP104   [990, -215]
1900-2000 change slit to 930
2000-0100(18 oct) JOP 137   [930, -470]

CDS
0000-0730 synoptic
0700-1000 JOP133  [850, -400]
1100-1930 JOP 104  [990, -215
2000-0140(18 oct) JOP 137  [925, -405

UVCS



1900-1600 (18 oct) long synoptics

TRACE request to support JOP 137,  JOP 104
YOHKOH support requested for JOP 137 and JOP 104

EIT
Normal CME watch is sufficient support


